Author's response to reviews

Title: The feasibility of a single-blinded fast-track pragmatic randomised controlled trial of a complex intervention for breathlessness in advanced disease.

Authors:

Morag C Farquhar (mcf22@medschl.cam.ac.uk)
Irene J Higginson (Irene.higginson@kcl.ac.uk)
Petrea Fagan (petrea.fagan@addenbrookes.nhs.uk)
Sara Booth (sara.booth@addenbrookes.nhs.uk)

Version: 3 Date: 29 June 2009

Author's response to reviews: see over
The Editor  
BMC Palliative Care

Dear Editor

Re: The feasibility of a single-blinded fast-track pragmatic randomised controlled trial of a complex intervention for breathlessness in advanced disease.

Thank you for accepting the above paper. The formatting requirements you requested have been made as follows:

1) Author list – this has been revised as requested.  
2) Boxes have been changed to tables or figures or incorporated into the text as requested.  
3) Box 3 (and Box 7) have been changed to figures.  
4) Table vertical elements – on telephoning your offices for clarification (28/06/09) I was advised that this correction can be made by your offices.  
5) Table 4 has now been made into an Additional File.  
6) Affiliations and email addresses are now presented as per your instructions.  
7) Table, box and figure placement text has been removed.  
8) Abbreviations are now a paragraph.  
9) Authors’ contributions are now a paragraph.

Yours sincerely

Dr Morag Farquhar  
Macmillan Post-Doctoral Research Fellow  
Email: mcf22@medschl.cam.ac.uk